Author's response to reviews

Title: Variations in rates of nosocomial infection among Canadian neonatal intensive care units may be practice-related

Authors:

Khalid Aziz (kaziz@mun.ca)
Douglas D McMillan (Douglas.McMillan@iwk.nshealth.ca)
Wayne Andrews (wandrews@morgan.ucsf.mun.ca)
Margaret Pendray (mpendray@cw.bc.ca)
Zhenguo Qiu (winston_qiu@yahoo.ca)
Stella Karuri (skaruri@cw.bc.ca)
Shoo K Lee (shool@interchange.ubc.ca)
Canadian NEONATAL Network (hbavinton@cw.bc.ca)

Version: 5 Date: 29 June 2005

Author's response to reviews:

We have revised the manuscript as per the formatting instructions of the editor.